

JGR Biogeosciences

RESEARCH ARTICLE

10.1029/2021JG006478

Key Points:

- Emissions of methane and carbon dioxide can be large from forest ditches on both mineral and peat soils
- Ditches are constructed and therefore these emissions are anthropogenic in origin and should be accounted for in inventories
- Ditch methane emissions are particularly important, as they can influence landscape-scale methane budgets

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

M. Peacock, michael.peacock@slu.se

Citation:

Peacock, M., Granath, G., Wallin, M. B., Högbom, L., & Futter, M. N. (2021). Significant emissions from forest drainage ditches—An unaccounted term in anthropogenic greenhouse gas inventories? *Journal of Geophysical Research: Biogeosciences*, *126*, e2021JG006478. https://doi. org/10.1029/2021JG006478

Received 10 JUN 2021 Accepted 6 SEP 2021

Author Contributions:

Conceptualization: Mike Peacock, Martyn N. Futter Formal analysis: Mike Peacock Funding acquisition: Mike Peacock, Martyn N. Futter Investigation: Mike Peacock, Gustaf Granath, Marcus B. Wallin, Lars Högbom Writing – original draft: Mike Peacock

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Significant Emissions From Forest Drainage Ditches—An Unaccounted Term in Anthropogenic Greenhouse Gas Inventories?

Mike Peacock¹, Gustaf Granath², Marcus B. Wallin¹, Lars Högbom^{3,4}, and Martyn N. Futter¹

¹Department of Aquatic Sciences and Assessment, Swedish University of Agricultural Sciences, Uppsala, Sweden,
²Department of Ecology and Genetics, Uppsala University, Uppsala, Sweden, ³Skogforsk, Uppsala, Sweden,
⁴Department of Forest Ecology and Management, Swedish University of Agricultural Sciences, Umeå, Sweden

Abstract Forestry is a natural climate solution for removing atmospheric carbon dioxide (CO₂) and reaching net zero emissions. Managed boreal forests typically have extensive drainage ditch networks, and these can be hotspots of anthropogenic greenhouse gas (GHG) emissions, potentially offsetting the terrestrial carbon gain. However, there is a lack of data detailing GHG emissions from ditches on mineral soils, where most boreal forestry occurs. Here, we address this knowledge gap using two approaches. First, we conducted a synoptic campaign to measure summer GHG fluxes from 109 boreal forest ditches draining mineral soils within one local region. We found a clear control of ditch water level on methane (CH_{4}) , with zero emissions from dry ditches and variable, but often high, emissions from water-filled ditches. Almost all ditches acted as sources of CO₂, regardless of water status. Second, we reanalyzed a published data set of boreal forest ditches and streams across three regions where GHG concentrations had been repeatedly measured and detailed catchment information was available. Within this data set we categorized 76 ditches into mineral and peatland catchments and detected no difference in mean CH_{4} and CO₂ concentrations between the two soil types. GHG emissions from ditches draining mineral forest soils can be as large as those from peatland forest ditches. Using literature values for forest GHG uptake we demonstrate that ditch CH₄ emissions are particularly important and can offset the terrestrial CH₄ uptake. Ignoring ditch emissions, which are anthropogenic in origin, will lead to incorrect estimates of the landscape-scale forest GHG budget.

Plain Language Summary Managed boreal forests can remove large amounts of carbon dioxide from the atmosphere. It is hoped that this removal will go some way to counteracting anthropogenic greenhouse gas (GHG) emissions. However, forests often contain networks of human-constructed drainage ditches which can be hotspots for emissions of methane, a powerful GHG. Most boreal forests are found on mineral soils but there is almost no information concerning the size of forest ditch GHG emissions from these soils. We measured GHGs in a large number of boreal forest ditches and found that emissions were of a similar average size from ditches draining mineral and peat soils. Dry ditches emitted no methane, but water-filled ditches emitted high amounts. All ditches, dry and wet, emitted carbon dioxide. We show that even though ditches occupy a small proportion of the forest area, their emissions can influence landscape-scale GHG budgets.

1. Introduction

"In the beginning there was a mire, a ditching mattock and Jussi"

The opening lines of the Finnish novel Under the North Star (published 1959) by Väinö Linna, in which the farm hand Jussi is draining a mire.

Drainage ditches are common features of many production landscapes. They are constructed for various reasons but are generally used to lower the terrestrial water table in order to improve agricultural and forest productivity, by increasing both vegetation rooting depth and nutrient availability via the decomposition of organic matter. There is a wide variety of ditch morphologies and characteristics; it is not unusual for



Writing – review & editing: Gustaf Granath, Marcus B. Wallin, Lars Högbom, Martyn N. Futter ditches to sometimes dry up, and these systems therefore occupy a changing interface between the aquatic and terrestrial realms. Perhaps this has led to ditches becoming a scientific "no man's land"? Certainly, they are extremely under-researched when compared to streams; their counterparts of natural origin (Koschorreck et al., 2020). Despite this, ditches are important components in many landscapes and often perform important ecosystem functions, for example, biodiversity provision (Williams et al., 2004), nutrient and pollutant retention (Herzon & Helenius, 2008), and stormwater removal (Needelman et al., 2007).

Ditches often have specific attributes (nutrient-rich sediments, anoxic waters, etc.) that result in the production and emission of the greenhouse gases (GHGs) methane (CH_4) and carbon dioxide (CO_2). Although ditches typically only occupy small fractions of an ecosystem's area, their emissions can be sizeable, and important on landscape scales for both CH_4 (Cooper et al., 2014; Peacock et al., 2017; Roulet & Moore, 1995; Teh et al., 2011) and CO_2 (Hyvönen et al., 2013; Jauhiainen & Silvennoinen, 2012; Xiao et al., 2021). Ditches, particularly those in intensive agricultural environments, can also emit nitrous oxide (N_2O) (Peacock et al., 2017; Teh et al., 2011; Webb et al., 2021), but these emissions are often lower than those from the adjacent terrestrial ecosystem (Evans et al., 2016). Across regional and global scales ditches further exert effects; a recent synthesis calculated that ditches emit 1% of all anthropogenic CH_4 (Peacock et al., 2021) and in countries with large areas of drained land (e.g., The Netherlands) it has been estimated that ditches are responsible for 16% of national emissions (Koschorreck et al., 2020). Furthermore, because ditches are human-constructed their emissions are classed as anthropogenic and need to be included in GHG inventories (IPCC, 2014, 2019).

The widespread adoption of the Paris Agreement, which aims to limit increases in global temperatures to 1.5° C, has fuelled interest in natural climate solutions (NCS). One promising NCS is forestry, which includes reforestation, avoided forest conversion, and improved forest management (Griscom et al., 2017). The possible benefits of forestry as a NCS could be compromised by climate-driven risks (e.g., drought, fire; Anderegg et al., 2020) but forestry has a central role in plans for limiting net GHG emissions in many countries. In Sweden, the government aims to reach net zero (of GHG emissions) by 2045, and CO₂ removal by forests will play a key role (Government Offices of Sweden, 2020). It has recently been argued that GHG emissions from wetlands, lakes, and streams offset ~50% of the Swedish land-use sink (Lindroth & Tranvik, 2021), but most of these emissions, under IPCC definitions, are natural in origin and therefore irrelevant for current IPCC accounting. However, forest ditch GHG emissions are anthropogenic and, if they substantially offset the forest carbon sink, they become highly relevant for national accounting.

Emissions of GHGs, particularly CH_{4} , from forest ditches are well represented in the literature (Koskinen et al., 2016; Roulet & Moore, 1995; Schiller & Hastie, 1996). In Finland, where 5.5 Mha of peatlands have been drained for forestry, total ditch CH_4 emissions have been calculated as 0.05 Tg, equivalent to 9% of the total national CH₄ emissions (Peacock et al., 2021). However, past studies have focused almost exclusively on ditches draining forests on mires/peatlands (typically defined as soils with peat depth \geq 30 cm) and in our recent CH_4 synthesis (Peacock et al., 2021) of all global ditch studies we identified only one study of forest ditches draining mineral soils (organic horizon <10 cm, or <20 cm and <12% organic carbon content, IPCC, 2014) where annual emissions could be derived (Klaus et al., 2018; although in some cases this may be an issue of semantics, with ditches being called streams in the literature). Generally, fluvial GHG emissions from low-order forest streams are driven by terrestrial production, either through heterotrophic or autotrophic respiration, and with CH₄ being produced in near-stream anoxic environments (Campeau et al., 2018, 2019). Lateral inputs of soil-derived dissolved organic carbon (DOC) with subsequent in situ mineralization are also suggested to contribute to fluvial GHGs (Evans et al., 2016). However, the breakdown of DOC is known to be related to water retention time (Catalán et al., 2016), and DOC can be adsorbed by mineral soils and then mineralized (Kothawala et al., 2009), so where in the landscape this GHG production occurs is system specific and largely related to hydrological conditions. It is possible that both the relative importance of external and internal production, as well as the magnitude of ditch GHG fluxes, may differ between peat and mineral soils. Thus, there is a clear knowledge gap concerning the "true" landscape-scale (i.e., including aquatic fluxes) GHG balance of managed forestry on mineral soils.

This knowledge gap is particularly relevant for Sweden where the majority of managed forests are on mineral soils. Ditching of these soils is commonplace (Ågren & Lidberg, 2019) and is conventionally done to either lead water through the landscape or to lower the water table in wetter areas (Jakobsson, 2013). However,





Figure 1. Mean daily temperature (red line) and rainfall (blue bars) for the sampling period, and sampling days (black circles).

ditching in Sweden has been practised for hundreds of years and its history has a convoluted path; for instance, forest drainage has at times taken place because it was thought to reduce frost, to prevent paludification, and because it was considered to be good for the environment (Jacks, 2019). The result of this is that ditch networks in the Swedish landscape are prevalent and extensive. A national analysis of Swedish streams <6 m wide (i.e., not just headwaters) found that, across all land cover types, ditches make up 63% of the network and 4% are straightened streams (Ågren & Lidberg, 2019). In forested headwater catchments the extent of stream network modification further increases; for example, in an individual forested Swedish catchment it was found that there were 149 km of ditches compared to a perennial stream length of 178 km, and that 22% of this perennial stream network had been straightened (Hasselquist et al., 2018).

We initiated a two-part study to investigate GHGs in boreal forest ditches. Firstly, to quantify the magnitude and spatial variation of GHG emissions from forest ditches on mineral soils, we surveyed 109 individual forest ditches in one locality in southern Sweden. We measured growing season fluxes of CH_4 and CO_2 , ditch morphology, and water chemistry. This survey measured GHG fluxes from ditches on mineral soils, but some sites may have peat soils upstream in the catchment that release organic matter and dissolved GHGs which are subsequently transported down the drainage network and emitted from areas of mineral soils (i.e., where we measured fluxes). Therefore, for the second part of our study, we compared the influence of mineral and peat soils on forest ditch GHG emissions by analyzing a subset of a published data set of CH_4 and CO_2 concentration measurements from ~300 fluvial forested headwater sites across three large regions of Sweden (Wallin et al., 2014, 2018). In the original publications these sites were described as "streams" but a significant proportion of forested headwater watercourses in Sweden are non-natural or modified (Kuglerová et al., 2017). We identified ditches within this data set and tested whether GHGs differed between ditches on mineral and peat soils.

2. Materials and Methods

2.1. Local Survey of Ditch GHG Fluxes

We sampled 109 forest ditches (Figures S1 and S2) during the 2020 growing season. The sampling took place on 9 days between 25th June and 21st August (Figure 2), between 9:00 and 16:20. The sampled sites were all within an area centered on the city of Uppsala (59°51′29″N 17°38′41″E) with an approximate radius of 40 km. The climate is humid continental, and for 2020 mean annual temperature was 8.9°C and total precipitation was 503 mm (SMHI data from "Uppsala Aut" station—https://www.smhi.se/). During the study period the mean daily temperature was 18.1°C and 101 mm of rain fell (Figure 1). The 1991–2020 daily mean temperatures are 18.2°C and 16.9°C for July and August respectively, and respective mean monthly rainfall is 57.4 and 73.6 mm. The region is dominated by boreal forest; for Uppsala County (where 100 of the ditches were located) 66% of the land area is forest (Statistics Sweden, 2019). The principal trees species are Norway spruce (*Picea abies*), Scots pine (*Pinus sylvestris*), and birches (*Betula pendula* and *Betula pubescens*), with a shrub layer dominated by blueberry (*Vaccinium myrtillus*) and lingonberry (*Vaccinium*



vitis-idaea), and a ground layer of red-stemmed feathermoss (*Pleurozium schreberi*), glittering woodmoss (*Hylocomium splendens*), and *Polytrichum* and *Cladonia* species. Aerial photography from the period 1955–1967 showed that 33 of the 109 ditches were originally on agricultural land, which has since been converted to Norway spruce forestry.

Ditches were chosen using a combination of local knowledge and mapping. Soil types were checked using Geological Survey of Sweden maps (apps.sgu.se/kartgenerator). In four cases, soils were mapped as deep peat, but field observations showed that this was not the case (e.g., rocks were present). The most common soil types were glacial clay (n = 37) and sandy till (n = 37; see Table S1 for a full list of soil types). At each ditch we measured ditch depth and width, and sediment temperature, and noted the presence or absence of vegetation. If water was present we measured water depth and temperature, made a visual assessment of whether the water was flowing, and took a water sample for analysis. Finally, we measured GHG emissions using the static/floating chamber method.

2.2. GHG Measurements

To measure GHG fluxes we used a chamber adapted from Bastviken et al. (2015) with a circular base (diameter 31.5 cm) and a total volume of 9.56 l. The chamber was covered in silver foil to reflect sunlight and minimize internal heating. In water-filled ditches, the chamber was deployed floating and allowed to drift. In dry ditches, the chamber was gently pushed down into the sediment to create a seal without creating chamber artifacts. The chamber was deployed for 5 min at each ditch and was connected in a closed loop to a Picarro GasScouter measuring real-time concentrations of CH₄ and CO₂. Diffusive fluxes were calculated using linear regression between GHG concentration and chamber deployment time and were corrected for ambient atmospheric pressure and air temperature. We did not set an R^2 cut-off but used the approach of Peacock et al. (2017); all nonsignificant (p > 0.05) regressions are classified as zero fluxes, but significant fluxes with low R^2 values are accepted as real fluxes. On nine occasions our chamber deployments caught ebullition events. These were evident in the data as abrupt increases in CH₄ concentration as bubbles entered the chamber. Where these occurred, an ebullitive flux was calculated as the change in CH₄ concentration over time. The ebullition event was not included in the calculation of diffusive flux; instead, the chamber was removed from the water to air it out, then the deployment was repeated to measure the diffusive flux. Ebullitive and diffusive fluxes were summed to calculate total flux; these total fluxes are given in the text of Section 3.1, but figures include only diffusive emissions.

2.3. Water Chemistry Measurements

Water samples were stored in the dark at 4° C before analysis. Samples were analyzed for pH and electrical conductivity (EC), and full UV-visible absorbance scans were measured using a 1 cm pathlength cuvette and an Avantes AvaLight DH-S-BAL light source, on samples filtered at 0.45 µm. We used absorbance measurements at 270 and 350 nm to calculate DOC concentrations using a published model (Carter et al., 2012; Tipping et al., 2009). Furthermore, we used absorbance at 400 nm as a measure of water color and also calculated the E2:E3 ratio (250:365 nm) which is a proxy for dissolved organic matter (DOM) composition and has been related to DOM aromaticity and molecular weight (Peuravuori & Pihlaja, 1997). Total phosphorus (TP) was analyzed by the SWEDAC-accredited Geochemical Laboratory at the Swedish University of Agricultural Science (SLU, 2020).

2.4. Multi-Regional Analysis of Ditch GHG Concentrations

For the second part of our study, we analyzed a published data set of CH_4 and CO_2 concentrations from ~300 fluvial headwaters. The data were collected from watercourses in three boreal or hemi-boreal regions: south east Sweden, central Sweden, and south west Sweden (Figure S1; these regions are SES, DAL, and LAVI in Wallin et al., 2014, 2018). Land use in the sampled catchments was dominated by managed forests, with no urban areas, and agricultural land cover <5%. Each watercourse was sampled three times: in spring, summer, and autumn. Measurements of GHGs were made using a headspace method and analyzed on a gas chromatograph. All samples were above the limit of detection for CO_2 (50 ppm), but for CH_4 some samples were below the limit of detection (1 ppm) and these were treated as zeroes in the analysis (see Wallin



et al., 2014, 2018 for more information). Although the sampled waterbodies were referred to as "streams" we assumed that many of the watercourses in this data set could be ditches, or straightened streams (and we note that others in Sweden have used the term "streams" to refer to what are forest ditches [e.g., Klaus et al., 2018]). We therefore used maps and satellite images to examine all sampled watercourses. We classified watercourses as ditches if they were perfectly straight, if they made unnaturally sharp turns (e.g., 90° turns), or if they were clearly part of ditch networks (i.e., numerous parallel watercourses, geometric drainage networks). This resulted in a total of 168 ditches.

We used data from these 168 ditches (Wallin et al., 2014, 2018), which includes catchment land cover, to test whether there was a difference in GHG concentrations between watercourses draining peat or mineral soil catchments. Firstly, we tested for correlations between catchment peatland area and GHG concentrations in all 168 ditches. Secondly, we designated all catchments with zero peatland area as mineral catchments (n = 61), and all catchments with peatland cover $\geq 20\%$ as catchments with peatland influence (n = 15). We calculated mean GHG concentrations for each catchment type (mineral and peat).

2.5. Statistical Analysis

We tested for differences in mean GHGs and other ditch variables (e.g., ditch depth, width) between categories (e.g., water-filled vs. dry ditches, mineral vs. peat soils, vegetated vs. unvegetated ditches) employing *t*-tests. Where our GHG variables displayed non-normal distributions and unequal variances between categories we used Welch's *t*-test (Fagerland & Sandvik, 2009; Kasuya, 2001). CH₄ data are frequently skewed, and therefore we also use median tests where appropriate. For the local GHG survey we used Spearman (rho) and Pearson (*r*) correlations to test for monotonic and linear relationships between GHGs and ditch characteristics/water chemistry, for water-filled ditches. For the multi-regional analysis of ditch GHG concentrations we used Spearman correlations to test for relationships between GHGs and catchment peatland area. All analyses were run in SPSS 26.

3. Results

3.1. Local Survey of Ditch GHG Fluxes

Of the 109 surveyed ditches, 51 contained water ("water-filled") and 58 were dry. Water-filled ditches were significantly deeper and wider than dry ditches, and sediments temperatures were lower (Table 1). Although the mean water depth was relatively shallow (9 cm), the range was large with water depths up to 1 m. Mean concentrations of both TP (221 μ g l⁻¹) and DOC (24 mg l⁻¹) were relatively high, as was mean E2:E3 (4.8); a measure of DOM composition.

Overall mean GHG fluxes were 33.9 (range -1.3 to 1,390) mg CH₄ m⁻² d⁻¹ and 6,016 (range -720 to 32,470) mg CO₂ m⁻² d⁻¹. For CH₄, fluxes from water-filled ditches were significantly larger than those from dry ditches (Figure 2), and 39 dry ditches acted as small sinks of CH₄ (mean uptake of 0.43 CH₄ m⁻² d⁻¹). Ebullition events were recorded at nine ditches, with a mean flux of 1,061 (range 3–3,880) mg CH₄ m⁻² d⁻¹, and ebullition comprised a mean of 75% (range 35%–99%) of total flux. There was no detectable difference in CO₂ fluxes between water-filled and dry ditches (Figure 3) and the majority of ditches were CO₂ sources; three ditches had zero emissions and five ditches were CO₂ sinks (mean uptake of 404 mg CO₂ m⁻² d⁻¹). More ditches were unvegetated (n = 63) than vegetated (n = 46), and we found no effect of vegetation on fluxes of CH₄ or CO₂ (see Table S2).

For water-filled ditches, CH_4 and CO_2 fluxes were significantly related to each other (Figure 4, Spearman correlation: rho = 0.57, p < 0.001). When considering water chemistry and ditch characteristics, positive correlations with GHG fluxes were detected for ditch water depth, but the relationships were weak (Figure 4, respective rho values of 0.4 and 0.29 for CH_4 and CO_2). There was no significant difference in the magnitude of GHG fluxes between flowing and standing ditches. However, there was a significant positive correlation between GHGs and DOC concentration for ditches with standing water (Figure 5, respective r values of 0.53 and 0.54 for CH_4 and CO_2). These correlations did not exist for ditches with flowing water, or for the combined data set of flowing and standing water.



Table 1

Means (±Standard Error of the Mean, SEM) and Ranges of Ditch Characteristics and Water Chemistry for 109 Sampled Forest Ditches, Grouped According to Whether Ditches Contained Water ("Water-Filled") or Were Dry

	Water-filled ($n = 51$)		Dry (n = 58)	
	Mean ± SEM	Range	Mean \pm SEM	Range
Ditch depth (cm)	118 ± 8	30-250	79 ± 5	20-220
Ditch width (cm)	165 ± 11	35-400	100 ± 6	50-290
Sediment temp (°C)	12.8 ± 1.0	10.4–15.7	14.0 ± 0.2	10.5-18.0
Water temp (°C)	13.7 ± 0.4	10.5-26.1	-	-
Water depth (cm)	9 ± 2	1–100	-	-
рН	6.89 ± 0.07	5.61-7.80	-	-
$EC (\mu S cm^{-1})$	157 ± 14.5	30-440	-	-
DOC (mg l ⁻¹)	24.2 ± 1.6	5.3-52.9	-	-
$TP (\mu g l^{-1})$	221.4 ± 43.5	6.9–1,240	-	-
Water color (400 nm)	0.133 ± 0.014	0.003-0.458	-	-
E2:E3	4.8 ± 0.1	3.1-7.6	-	-

Note. n = 35 for sediment temp in water-filled ditches because deep water prevented temperature measurements being taken in 16 ditches. There are significant ($p \le 0.001$) differences between water-filled and dry ditches for the first three variables. DOM, dissolved organic carbon; EC, electrical conductivity; TP, total phosphorus.

3.2. Multi-Regional Analysis of Ditch GHG Concentrations

Analysis of the published data set of ditch GHG concentrations revealed no significant difference in mean CH_4 or CO_2 concentrations between ditches draining catchments on mineral or peat soil, but median concentrations of CH_4 were significantly greater in peat catchments (Figure 6). There was a very weak but significant correlation between catchment peat cover and CH_4 concentration (rho = 0.18, p = 0.02), but no relationship with CO_2 (Figure 7).

4. Discussion

4.1. Local Survey of Ditch GHG Fluxes

Our synoptic survey of GHG fluxes from 109 forest ditches revealed an overriding control of ditch water status on CH₄ emissions. The mean emission from dry ditches was zero, and CH₄ uptake was often observed. In contrast to this, relatively large emissions were observed from water-filled ditches, with a mean flux of 72 mg CH₄ m⁻² d⁻¹. For water-filled ditches, we found a weak positive relationship between CH₄ flux and water depth. For other small waterbodies, such as peatland pools, negative relationships have been found (McEnroe et al., 2009), and this has been attributed to deeper waters being colder and suppressing methanogenesis (Pelletier et al., 2007). However, as suggested elsewhere (Peacock et al., 2017), we assume that ditches with deeper water levels during summer (when we sampled) are likely to have reduced water level fluctuations, and less likely to dry out entirely, which will result in both the development of anoxia and the establishment of microbial communities favorable to CH_4 production. As in other studies of fluvial systems (Campeau & Del Giorgio, 2014; Wallin et al., 2014), CH₄ and CO₂ were positively correlated in our water-filled ditches perhaps because, to some extent, methanogenesis and ecosystem respiration are driven by the same environmental factors (Stanley et al., 2016). It is also feasible that anaerobic metabolism could be a shared source, resulting in oversaturation of CH₄ and CO₅ (Wallin et al., 2018). In contrast to CH₄, our CO, fluxes did not differ between water-filled and dry ditches but, like CH_4 , we observed a positive correlation between water depth and CO₂ flux. We speculate that this could be because a greater depth indicates a stronger soil-water connection, which results in enhanced lateral inputs of carbon. Intriguingly, we found positive relationships between DOC concentration and both GHGs but only for ditches with standing, rather than flowing, water. This suggests that, due to the increased water residence time, in situ breakdown of DOC contributes to emissions of CO₂ and CH₄ in these standing water bodies.





Figure 2. Box plots of diffusive CH_4 fluxes from dry (n = 58) and water-filled (n = 51) ditches, visualized three ways: (a) all data including outliers; (b) outliers not shown for water-filled ditches; (c) all data, log10 transformed, including outliers. Boxes represent medians and interquartile range (IQR), whiskers mark minimum and maximum values, excluding outliers (calculated as box limits $\pm 1.5 \times IQR$). Also shown are mean fluxes (x) and outliers (o). CH_4 fluxes are significantly different between categories (Welch's *t*-test).

We compared our CH_4 and CO_2 fluxes to other studies of forest ditches (Figure 8). For CH_4 , the majority of reported data are from mires and peatlands, but our fluxes are of the same magnitude as most of these. We were only able to find three other studies reporting CO₂ fluxes from forest ditches (on any soil type) but, again, these fall within the same magnitude as our measurements. Assuming a 3-month summer period and upscaling accordingly (i.e., the same flux over the whole period) would give summer emissions of 6.5 g CH_4 m⁻² and 540 g CO_2 m⁻². The annual emissions are likely to be considerably higher, because spring and autumn fluxes can account for a significant proportion of the total flux (Peacock et al., 2017) and winter emissions can be high, even for boreal forest ditches (Minkkinen & Laine, 2006). Nevertheless, the total annual emissions are likely to be lower than those from ditches in more intensively managed landscapes such as grasslands and croplands (Evans et al., 2016). We also acknowledge that there is potential for photosynthetic CO, uptake in some of our ditches during the growing season; 60% were unvegetated, but 40% were vegetated to some extent. Von Arnold et al. (2005) used transparent chambers on ditches in a drained spruce forest on mineral soils, and measured a net efflux of CO, during the growing season, but the combined use of dark and light chambers would allow a full quantification of photosynthetic uptake versus respiratory emission. We found no effect of vegetation on ditch GHG fluxes but this may be due to the basic nature of our assessment which simply categorized the presence or absence of vegetation. More detailed observations of vegetation type and percentage cover would likely reveal some





linkages with GHGs, especially with sedges which have been shown to enhance ditch CH_4 fluxes, either by supplying labile substrates for methanogenesis or by providing a direct conduit for atmospheric CH_4 emission (Cooper et al., 2014; Minkkinen & Laine, 2006).

At nine sites our chamber captured ebullition events which, on average, contributed 75% of the total CH_4 flux. This is high compared to the only other report of forest ditch ebullition we are aware of, where CH_4 emissions were dominated by the diffusive pathway (Minkkinen et al., 1997). Data from ditches in other ecosystems provide conflicting evidence, with ebullition being cited as being an important component of total flux (Panneer Selvam et al., 2014; Vermaat et al., 2011) or being negligible (Green et al., 2016; Köhn et al., 2021). The high ebullitive fluxes in our ditches could be due to a nutrient effect (Davidson et al., 2018), as total P concentrations were relatively high.

4.2. Ditches on Mineral Versus Peatland Soils

A comparison with other forest ditch GHG measurements (Figure 8), the majority of which have been taken from peatlands and mires, at least for CH_4 , suggests that ditches on mineral soils can emit similar





Figure 4. Scatter plots of greenhouse gases and ditch water depth for all water-filled ditches (n = 51). Note that CH₄ was transformed by adding 2 to all fluxes so all data points could be displayed on a log axis, but statistical tests were performed on untransformed data.

amounts of GHGs to ditches on peat soils. We further tested this with our multi-regional analysis of GHG concentrations from forest ditches that were repeatedly sampled during spring, summer, and autumn. No significant differences were found in mean CH_4 or CO_2 between ditches draining catchments with exclusively mineral soils, compared to ones that had $\geq 20\%$ peat cover, but median CH_4 concentrations were significantly greater in peat catchments by a factor of three, and we detected a weak relationship between catchment peat cover and CH_4 . The mean CH_4 concentration for mineral soil catchments is beyond the interquartile range (Figure 6) and this (along with the same finding for the CH_4 fluxes in Figure 2) revels the skewed nature of the CH_4 concentration and flux data. The skewed nature of CH_4 data is well known and brings forth the issue of whether to use the median or mean for upscaling and comparisons (Rosentreter & Williamson, 2020). However, assuming that sampling efforts have not been biased toward measuring GHG hotspots, these high emitting sites are representative of the entire population and thus the mean is the better measure (Al-Haj & Fulweiler, 2020; Paneer Selvam et al., 2014). Indeed, our data demonstrate the importance of large-scale, spatially replicated sampling, in order to capture these rare high- CH_4 emitting sites that are important to fully evaluate landscape-scale budgets.

What, then, are the mechanisms for the differences and similarities in GHGs between catchment soil types? First, DOC concentrations have been linked to fluvial CO_2 and CH_4 concentrations (Rasilo et al., 2017; Stanley et al., 2016; Wallin et al., 2014), suggesting a common spatial origin in the landscape of the different



Figure 5. Scatter plots of greenhouse gases and dissolved organic carbon (DOC) concentration, grouped by ditches with standing water (large, black-filled circles, n = 28) and flowing water (smaller, white-filled circles, n = 23). Pearson correlations are significant for ditches with standing water (for CH₄, r = 0.53, p = 0.004; for CO₂, r = 0.54, p = 0.003), but not for ditches with flowing water (for CH₄, r = -0.08, p = 0.7; for CO₂, r = -0.3, p = 0.16), or the combined data set of standing + flowing water (for CH₄, r = 0.19, p = 0.17; for CO₂, r = 0.27, p = 0.052). Note that CH₄ was transformed by adding 2 to all fluxes so all data points could be displayed on a log axis, but statistical tests were performed on untransformed data.





Figure 6. Box plots of dissolved greenhouse gas concentrations in ditches from the multi-regional analysis, grouped according to catchment soil type classification (mineral = 0% peat soil n = 61, peat = $\geq 20\%$ peat soil n = 15. Panels (a and c) show all CH₄ and CO₂ data including outliers; panels (b and d) show the same data with no outliers shown. Boxes represent medians and interquartile range (IQR), whiskers mark minimum and maximum values, excluding outliers (calculated as box limits $\pm 1.5 \times IQR$). Also shown are mean fluxes (x) and outliers (o). CO₂ concentrations between categories are not significant (*t*-test). Mean CH₄ concentrations between categories are not significant (*t*-test) but medians are (median test).

carbon species, although not necessarily from a common metabolic source (Campeau et al., 2019). In-keeping with the similarity of GHG concentrations between mineral and peat catchments, DOC concentrations in our multi-regional analysis were also the same in ditches draining peat and mineral catchments (means of ~30 mg l⁻¹) and concentrations in our mineral ditches sampled during the GHG flux survey were also similar (mean = 24 mg l⁻¹). Thus, it appears that, regardless of soil type, lateral inputs of terrestrially derived carbon in shallow groundwater can sustain GHG and DOC concentrations in these ecosystems (Cam-



Figure 7. Scatter plots of greenhouse gas concentrations and catchment peatland area (n = 168). The relationship for CH₄ is not dependent on the value of ~200 µg C l⁻¹ at 57% peatland cover: removing this datapoint results in rho = 0.17, p = 0.033.





Figure 8. A literature summary of ditch CH_4 (top panel) and CO_2 (lower panel) emissions from ditches in temperate and boreal forests. All studies are on peatland soils unless labeled otherwise. Emissions are color-coded by study period (growing season only, or growing season + autumn/winter measurements). References and additional information (climate zone, measurement period) are in Table S3.

peau et al., 2018). We assume that some DOC removal may occur via mineral soil adsorption but that carbon inputs from small riparian areas, thin organic layers, or even unmapped pockets of peat soils override any adsorption (Leith et al., 2015). When comparing medians and when expressed on a continuous scale $(CH_4 \text{ vs. }\% \text{ peat cover})$, the tendency for catchments with more peat cover to have higher CH_4 concentrations is likely an effect of increased lateral inputs from anaerobic soils (Minkkinen & Laine, 2006; Rasilo et al., 2017), which can potentially be sizable even in drained forest peats (Roulet & Moore, 1995). There is also the opportunity for in situ production within the ditch sediments which is likely to be greater in the organic-rich sediments of peatland ditches (Roulet & Moore, 1995). For catchments with greater areas of peat, most of the CH₄ produced in situ and via lateral inputs will likely be emitted rapidly, and to a lesser extent transported down the drainage network. Another possibility is that a difference in DOM composition is implicated; mean E2:E3 in our water-filled mineral ditches was 4.8, considerably higher than values from peatland ditches (3-3.5; Peacock et al., 2018; Strack et al., 2015). In a study of two hemi-boreal headwater forest catchments, Wallin et al. (2015) also found a significantly higher E2:E3 in a stream draining a mineral catchment compared to an adjacent peatland catchment. A lower E2:E3 is indicative of higher aromaticity (Peuravuori & Pihlaja, 1997), which has been shown to be associated with elevated concentrations of aquatic CH_4 (Zhou et al., 2018).



4.3. Implications

Our results show that GHG emissions from ditches on mineral soils in boreal forests are non-negligible. The 2013 IPCC Wetlands Supplement (IPCC, 2014) highlighted peatland drainage ditches as being important sources of CH₄ emission that could influence landscape-scale GHG budgets (Evans et al., 2016) but assumed that drained wetlands on mineral soils had zero CH₄ emissions; that is, it did not consider ditches. Regardless of the pre-drainage status of the soils, the ditches should now be considered for inclusion when calculating the landscape-scale forest GHG balance. To test this, we took values for GHG uptake in boreal forests on mineral soils as 0.52 mg CH₄ m⁻² d⁻¹ (the mean of boreal forest mineral soil fluxes from Dutaur & Verchot, 2007) and 15,200 mg CO₂ m⁻² d⁻¹ (the mean of whole-forest peak summer uptake in Swedish forests, from Chi et al., 2020; Lindroth et al., 1998). We then took our mean ditch CO, flux, and our mean CH₄ flux from water-filled ditches only, and calculated what fraction of the landscape water-filled ditches would have to occupy ("Frac_{ditch}"; Evans et al., 2016) for their GHG emissions to offset the terrestrial forest uptake. For CH_4 , the calculated $Frac_{ditch}$ is 0.007 (0.7%) while for CO_2 $Frac_{ditch}$ is 0.72 (72%). The majority of the calculated $Frac_{ditch}$ is 0.007 (0.7%) while for CO_2 $Frac_{ditch}$ is 0.72 (72%). Frac_{ditch} values, for both peat and mineral soils, are much higher than 0.007, although there are no reported Frac_{ditch} values for forests on mineral soils (Peacock et al., 2021). Care is needed when generalizing our results as they are from summer sampling in one region only. Nevertheless, they suggest, in keeping with research from peatland ditches (Evans et al., 2016), that mineral ditches act as landscape-scale hotspots for CH_4 emission but not CO_2 . However, this only holds true for CH_4 when ditches are water-filled. It may therefore be possible to incorporate ditch CH_4 emissions into inventories using simple modeling of water levels (Tucker & Acreman, 2000) or by direct measurements with low-cost sensors (Chapin et al., 2014). These data could be combined with emission factors (which would be zero for dry ditches) for CH₄ accounting, although we note that the current IPCC emission factor for mineral ditches (41.6 g CH, m^{-2} yr⁻¹) appears to be inappropriately high when compared to our summer emissions (6.5 g CH_4 m⁻² for the 3-month period). This is due to a widespread lack of flux data from non-peatland ditches resulting in a single generic emission factor for mineral ditches, regardless of climate zone or nutrient status (IPCC, 2019). Thus, further studies of mineral ditch CH₄, including comprehensive measurements of ebullition, are clearly needed to refine emission factors and enable accurate GHG accounting for managed lands. Future climatic changes in boreal northern Europe are likely to lead to lower amounts of summer rainfall, more frequent droughts, and therefore low water availability (Ruiz-Pérez & Vico, 2020). These changes may have the effect of mitigating growing season ditch CH₄ emissions if watercourses dry out on a regular basis, but higher temperatures may also enhance ditch CO₂ emissions. Furthermore, changes to the timing of precipitation events could increase the occurrence of drought-rewetting cycles which can be hot moments for fluvial GHG emission (Arce et al., 2021; Wallin et al., 2020). Regardless of future changes, it is clear that ditches have the potential to offset, at least to some extent, the NCS of GHG uptake provided by drained forested landscapes.

Data Availability Statement

The data from the survey of ditch GHG fluxes are available in Table S3 in Peacock (2021, https://figshare. com/articles/dataset/Forest_ditch_GHG_data/15152253). The full data (i.e., from all streams) from the multi-regional sampling of ditch GHG concentrations are available at the Uppsala University data repository: http://urn.kb.se/resolve?urn=urn:nbn:se:uu:diva-332472. The specific data analyzed here (i.e., just the watercourses identified as ditches) are available in Table S4 in Peacock (2021, https://figshare.com/articles/dataset/Forest_ditch_GHG_data/15152253).

References

Ågren, A. M., & Lidberg, W. (2019). The importance of better mapping of stream networks using high resolution digital elevation models-upscaling from watershed scale to regional and national scales. *Hydrology and Earth System Sciences Discussions*. https://doi. org/10.5194/hess-2019-34

Arce, M. I., Bengtsson, M. M., von, Schiller, D., Zak, D., Täumer, J., Urich, T., & Singer, G. (2021). Desiccation time and rainfall control gaseous carbon fluxes in an intermittent stream. *Biogeochemistry*, 155, 381–400. https://doi.org/10.1007/s10533-021-00831-6

Acknowledgments

This research was funded by SLU's Environmental Monitoring and Assessment Climate Programme, in a grant awarded to Mike Peacock and Martyn N. Futter (project: Quantifying the magnitude of greenhouse gas emissions from forest ditches). Mike Peacock also acknowledges funding from the Swedish Research Council Formas (Grant 2020-00950). Marcus B. Wallin acknowledge funding from Formas (Grant 2019-01105). The authors thank the Geochemical Laboratory at the Swedish University of Agricultural Sciences for total P analysis. The authors thank Skogssällskapet for allowing sampling access to their Bredvik site. The authors thank Tim Moore and Corianne Tatariw for their constructive comments that helped to improve the manuscript.

Al-Haj, A. N., & Fulweiler, R. W. (2020). Response to concerns and uncertainties relating to methane emissions synthesis for vegetated coastal ecosystems. *Global Change Biology*, *26*, e10–e11.

Anderegg, W. R., Trugman, A. T., Badgley, G., Anderson, C. M., Bartuska, A., Ciais, P., et al. (2020). Climate-driven risks to the climate mitigation potential of forests. *Science*, 368(6497). https://doi.org/10.1126/science.aaz7005



- Ball, T. O. M., Smith, K. A., & Moncrieff, J. B. (2007). Effect of stand age on greenhouse gas fluxes from a Sitka spruce [Picea sitchensis (Bong.) Carr.] chronosequence on a peaty gley soil. *Global Change Biology*, *13*, 2128–2142. https://doi.org/10.1111/j.1365-2486.2007.01427.x
- Bastviken, D., Sundgren, I., Natchimuthu, S., Reyier, H., & Gålfalk, M. (2015). Cost-efficient approaches to measure carbon dioxide (CO₂) fluxes and concentrations in terrestrial and aquatic environments using mini loggers. *Biogeosciences*, *12*(12), 3849–3859. https://doi. org/10.5194/bg-12-3849-2015
- Campeau, A., Bishop, K., Amvrosiadi, N., Billett, M. F., Garnett, M. H., Laudon, H., et al. (2019). Current forest carbon fixation fuels stream CO₂ emissions. *Nature Communications*, *10*, 1876. https://doi.org/10.1038/s41467-019-09922-3
- Campeau, A., Bishop, K., Nilsson, M. B., Klemedtsson, L., Laudon, H., Leith, F. I., et al. (2018). Stable carbon isotopes reveal soil-stream DIC linkages in contrasting headwater catchments. *Journal of Geophysical Research: Biogeosciences*, 123(1), 149–167. https://doi. org/10.1002/2017jg004083
- Campeau, A., & Del Giorgio, P. A. (2014). Patterns in CH₄ and CO₂ concentrations across boreal rivers: Major drivers and implications for fluvial greenhouse emissions under climate change scenarios. *Global Change Biology*, 20, 1075–1088. https://doi.org/10.1111/gcb.12479 Carter, H. T., Tipping, E., Koprivnjak, J. F., Miller, M. P., Cookson, B., & Hamilton-Taylor, J. (2012). Freshwater DOM quantity and quality
- from a two-component model of UV absorbance. *Water Research*, 46(14), 4532–4542. https://doi.org/10.1016/j.watres.2012.05.021
- Catalán, N., Marcé, R., Kothawala, D. N., & Tranvik, L. J. (2016). Organic carbon decomposition rates controlled by water retention time across inland waters. *Nature Geoscience*, 9(7), 501–504. https://doi.org/10.1038/ngeo2720
- Chapin, T. P., Todd, A. S., & Zeigler, M. P. (2014). Robust, low-cost data loggers for stream temperature, flow intermittency, and relative conductivity monitoring. *Water Resources Research*, 50(8), 6542–6548. https://doi.org/10.1002/2013wr015158
- Chi, J., Nilsson, M. B., Laudon, H., Lindroth, A., Wallerman, J., Fransson, J. E., et al. (2020). The net landscape carbon balance—Integrating terrestrial and aquatic carbon fluxes in a managed boreal forest landscape in Sweden. *Global Change Biology*, 26, 2353–2367. https:// doi.org/10.1111/gcb.14983
- Cooper, M. D., Evans, C. D., Zielinski, P., Levy, P. E., Gray, A., Peacock, M., et al. (2014). Infilled ditches are hotspots of landscape methane flux following peatland re-wetting. *Ecosystems*, *17*(7), 1227–1241. https://doi.org/10.1007/s10021-014-9791-3
- Davidson, T. A., Audet, J., Jeppesen, E., Landkildehus, F., Lauridsen, T. L., Søndergaard, M., & Syväranta, J. (2018). Synergy between nutrients and warming enhances methane ebullition from experimental lakes. *Nature Climate Change*, *8*, 156–160. https://doi.org/10.1038/ s41558-017-0063-z
- Dutaur, L., & Verchot, L. V. (2007). A global inventory of the soil CH₄ sink. Global Biogeochemical Cycles, 21(4). https://doi. org/10.1029/2006gb002734
- Evans, C. D., Renou-Wilson, F., & Strack, M. (2016). The role of waterborne carbon in the greenhouse gas balance of drained and re-wetted peatlands. Aquatic Sciences, 78(3), 573–590. https://doi.org/10.1007/s00027-015-0447-y
- Fagerland, M. W., & Sandvik, L. (2009). Performance of five two-sample location tests for skewed distributions with unequal variances. Contemporary Clinical Trials, 30(5), 490–496. https://doi.org/10.1016/j.cct.2009.06.007
- Glagolev, M. V., Chistotin, M. V., Shnyrev, N. A., & Sirin, A. A. (2008). The emission of carbon dioxide and methane from drained peatlands changed by economic use and from natural mires during the summer-fall period (on example of a region of Tomsk oblast). *Agrokhimija*, *5*, 46–58.
- Government Offices of Sweden. (2020). Sweden's long-term strategy for reducing greenhouse gas emissions. Ministry of the Environment. Retrieved from https://unfccc.int/documents/267243
- Green, S. M., Baird, A. J., Evans, C., Peacock, M., Holden, J., Chapman, P., et al. (2016). The effect of drainage ditch blocking on the CO₂ and CH₄ budgets of blanket peatland. Appendix D. In S. M. Green, A. J. Baird, C. Evans, N. Ostle, J. Holden, P. J. Chapman, & N. McNamara (Eds.), *Investigation of peatland restoration (grip blocking) techniques to achieve best outcomes for methane and greenhouse gas emissions/balance: Field trials and process experiments final report* (Defra Project SP1202). University of Leeds.
- Griscom, B. W., Adams, J., Ellis, P. W., Houghton, R. A., Lomax, G., Miteva, D. A., et al. (2017). Natural climate solutions. Proceedings of the National Academy of Sciences, 114(44), 11645–11650. https://doi.org/10.1073/pnas.1710465114
- Hasselquist, E. M., Lidberg, W., Sponseller, R. A., Ågren, A., & Laudon, H. (2018). Identifying and assessing the potential hydrological function of past artificial forest drainage. *Ambio*, 47(5), 546–556. https://doi.org/10.1007/s13280-017-0984-9
- Herzon, I., & Helenius, J. (2008). Agricultural drainage ditches, their biological importance and functioning. *Biological Conservation*, 141(5), 1171–1183. https://doi.org/10.1016/j.biocon.2008.03.005
- Hyvönen, N. P., Huttunen, J. T., Shurpali, N. J., Lind, S. E., Marushchak, M. E., Heitto, L., & Martikainen, P. J. (2013). The role of drainage ditches in greenhouse gas emissions and surface leaching losses from a cutaway peatland cultivated with a perennial bioenergy crop. *Boreal Environment Research*, 18, 109–126.
- IPCC. (2014). Wetlands. In T. Hiraishi, T. Krug, K. Tanabe, N. Srivastava, J. Baasansuren, M. Fukuda, & T. G. Troxler (Eds.), 2013 supplement to the 2006 IPCC guidelines for national greenhouse gas inventories. IPCC.
- IPCC. (2019). Chapter 7: Wetlands. In Agriculture, forestry and other land use (Vol. 4). 2019 refinement to the 2006 IPCC guidelines for national greenhouse gas inventories.
- Jacks, G. (2019). Drainage in Sweden—The past and new developments. Acta Agriculturae Scandinavica Section B: Soil and Plant Science, 69(5), 405–410. https://doi.org/10.1080/09064710.2019.1586991
- Jakobsson, E. (2013). Ditching from a water system perspective. Draining the Swedish water landscape 1200–1900. Water History, 5(3), 349–367. https://doi.org/10.1007/s12685-013-0089-8
- Jauhiainen, J., & Silvennoinen, H. (2012). Diffusion GHG fluxes at tropical peatland drainage canal water surfaces. Suo, 63(3–4), 93–105. Kasuya, E. (2001). Mann-Whitney U test when variances are unequal. Animal Behaviour, 61, 1247–1249. https://doi.org/10.1006/ anbe.2001.1691
- Klaus, M., Geibrink, E., Jonsson, A., Bergström, A. K., Bastviken, D., Laudon, H., et al. (2018). Greenhouse gas emissions from boreal inland waters unchanged after forest harvesting. *Biogeosciences*, 15(18), 5575–5594. https://doi.org/10.5194/bg-15-5575-2018
- Köhn, D., Welpelo, C., Günther, A., & Jurasinski, G. (2021). Drainage ditches contribute considerably to the CH₄ budget of a drained and a rewetted temperate fen (version 1). https://doi.org/10.21203/rs.3.rs-223145/v1
- Korkiakoski, M., Ojanen, P., Penttilä, T., Minkkinen, K., Sarkkola, S., Rainne, J., et al. (2020). Impact of partial harvest on CH_4 and N_2O balances of a drained boreal peatland forest. *Agricultural and Forest Meteorology*, 295, 108168. https://doi.org/10.1016/j. agrformet.2020.108168
- Koschorreck, M., Downing, A. S., Hejzlar, J., Marcé, R., Laas, A., Arndt, W. G., et al. (2020). Hidden treasures: Human-made aquatic ecosystems harbour unexplored opportunities. *Ambio*, 49(2), 531–540. https://doi.org/10.1007/s13280-019-01199-6
- Koskinen, M., Maanavilja, L. M., Nieminen, M., Minkkinen, K., & Tuittila, E. S. (2016). High methane emissions from restored Norway spruce swamps in southern Finland over one growing season. *Mires & Peat*, 17, 2.



Kothawala, D. N., Moore, T. R., & Hendershot, W. H. (2009). Soil properties controlling the adsorption of dissolved organic carbon to mineral soils. Soil Science Society of America Journal, 73(6), 1831–1842. https://doi.org/10.2136/sssaj2008.0254

- Kuglerová, L., Hasselquist, E. M., Richardson, J. S., Sponseller, R. A., Kreutzweiser, D. P., & Laudon, H. (2017). Management perspectives on Aqua incognita: Connectivity and cumulative effects of small natural and artificial streams in boreal forests. *Hydrological Processes*, 31(23), 4238–4244. https://doi.org/10.1002/hyp.11281
- Leith, F. I., Dinsmore, K. J., Wallin, M. B., Billett, M. F., Heal, K. V., Laudon, H., et al. (2015). Carbon dioxide transport across the hillslope-riparian-stream continuum in a boreal headwater catchment. *Biogeosciences*, 12(6), 1881–1892. https://doi.org/10.5194/bg-12-1881-2015
- Lindroth, A., Grelle, A., & Morén, A. S. (1998). Long-term measurements of boreal forest carbon balance reveal large temperature sensitivity. *Global Change Biology*, *4*, 443–450. https://doi.org/10.1046/j.1365-2486.1998.00165.x
- Lindroth, A., & Tranvik, L. (2021). Accounting for all territorial emissions and sinks is important for development of climate mitigation policies. *Carbon Balance and Management*, 16(1), 1–3. https://doi.org/10.1186/s13021-021-00173-8
- McEnroe, N. A., Roulet, N. T., Moore, T. R., & Garneau, M. (2009). Do pool surface area and depth control CO₂ and CH₄ fluxes from an ombrotrophic raised bog, James Bay, Canada? *Journal of Geophysical Research*, *114*(G1). https://doi.org/10.1029/2007jg000639
- Minkkinen, K., & Laine, J. (2006). Vegetation heterogeneity and ditches create spatial variability in methane fluxes from peatlands drained for forestry. *Plant and Soil*, 285, 289–304. https://doi.org/10.1007/s11104-006-9016-4
- Minkkinen, K., Laine, J., Nykänen, H., & Martikainen, P. J. (1997). Importance of drainage ditches in emissions of methane from mires drained for forestry. *Canadian Journal of Forest Research*, 27, 949–952. https://doi.org/10.1139/x97-016
- Minkkinen, K., Ojanen, P., Penttilä, T., Aurela, M., Laurila, T., Tuovinen, J. P., & Lohila, A. (2018). Persistent carbon sink at a boreal drained bog forest. *Biogeosciences*, 15, 3603–3624. https://doi.org/10.5194/bg-15-3603-2018
- Needelman, B. A., Kleinman, P. J., Strock, J. S., & Allen, A. L. (2007). Drainage ditches: Improved management of agricultural drainage ditches for water quality protection: An overview. Journal of Soil and Water Conservation, 62(4), 171–178.
- Panneer Selvam, B., Natchimuthu, S., Arunachalam, L., & Bastviken, D. (2014). Methane and carbon dioxide emissions from inland waters in India–implications for large scale greenhouse gas balances. *Global Change Biology*, 20, 3397–3407. https://doi.org/10.1111/gcb.12575 Peacock, M. (2021). Forest ditch GHG data (Dataset). Figshare. https://doi.org/10.6084/m9.figshare.15152253.v1
- Peacock, M., Audet, J., Bastviken, D., Futter, M. N., Gauci, V., Grinham, A. R., et al. (2021). Global importance of methane emissions from drainage ditches and canals. *Environmental Research Letters*, 16, 044010. https://doi.org/10.1088/1748-9326/abeb36
- Peacock, M., Jones, T. G., Futter, M. N., Freeman, C., Gough, R., Baird, A. J., et al. (2018). Peatland ditch blocking has no effect on dissolved organic matter (DOM) quality. *Hydrological Processes*, 32(26), 3891–3906. https://doi.org/10.1002/hyp.13297
- Peacock, M., Ridley, L. M., Evans, C. D., & Gauci, V. (2017). Management effects on greenhouse gas dynamics in fen ditches. Science of the Total Environment, 578, 601–612. https://doi.org/10.1016/j.scitotenv.2016.11.005
- Pelletier, L., Moore, T. R., Roulet, N. T., Garneau, M., & Beaulieu-Audy, V. (2007). Methane fluxes from three peatlands in the La Grande Riviere watershed, James Bay lowland, Canada. *Journal of Geophysical Research*, 112(G1). https://doi.org/10.1029/2006jg000216
- Peuravuori, J., & Pihlaja, K. (1997). Molecular size distribution and spectroscopic properties of aquatic humic substances. Analytica Chimica Acta, 337, 133–149. https://doi.org/10.1016/s0003-2670(96)00412-6
- Rasilo, T., Hutchins, R. H., Ruiz-González, C., & Del Giorgio, P. A. (2017). Transport and transformation of soil-derived CO₂, CH₄ and DOC sustain CO₂ supersaturation in small boreal streams. *Science of the Total Environment*, 579, 902–912. https://doi.org/10.1016/j. scitotenv.2016.10.187
- Rosentreter, J. A., & Williamson, P. (2020). Concerns and uncertainties relating to methane emissions synthesis for vegetated coastal ecosystems. Global Change Biology, 26, 5351–5352. https://doi.org/10.1111/gcb.15201
- Roulet, N. T., & Moore, T. R. (1995). The effect of forestry drainage practices on the emission of methane from northern peatlands. Canadian Journal of Forest Research, 25(3), 491–499. https://doi.org/10.1139/x95-055
- Ruiz-Pérez, G., & Vico, G. (2020). Effects of temperature and water availability on Northern European boreal forests. Frontiers in Forests and Global Change, 3, 34. https://doi.org/10.3389/ffgc.2020.00034
- Schiller, C. L., & Hastie, D. R. (1996). Nitrous oxide and methane fluxes from perturbed and unperturbed boreal forest sites in northern Ontario. Journal of Geophysical Research, 101, 22767–22774. https://doi.org/10.1029/96jd01620
- Sirin, A. A., Suvorov, G. G., Chistotin, M. V., & Glagolev, M. V. (2012). Values of methane emission from drainage ditches. Environmental Dynamics and Global Climate Change, 3(2), 1–10. https://doi.org/10.17816/edgcc321-10
- SLU. (2020). Geochemical laboratory-accredited analytical methods. Retrieved from https://www.slu.se/en/departments/ aquatic-sciences-assessment/laboratories/vattenlabb2/
- Stanley, E. H., Casson, N. J., Christel, S. T., Crawford, J. T., Loken, L. C., & Oliver, S. K. (2016). The ecology of methane in streams and rivers: Patterns, controls, and global significance. *Ecological Monographs*, 86, 146–171. https://doi.org/10.1890/15-1027
- Statistics Sweden. (2019). Land use in Sweden 2015. Retrieved from https://www.scb.se/en/finding-statistics/statistics-by-subject-area/ environment/land-use/land-use-in-sweden/pong/tables-and-graphs/land-use-in-sweden-2015
- Strack, M., Zuback, Y., McCarter, C., & Price, J. (2015). Changes in dissolved organic carbon quality in soils and discharge 10 years after peatland restoration. *Journal of Hydrology*, 527, 345–354. https://doi.org/10.1016/j.jhydrol.2015.04.061
- Teh, Y. A., Silver, W. L., Sonnentag, O., Detto, M., Kelly, M., & Baldocchi, D. D. (2011). Large greenhouse gas emissions from a temperate peatland pasture. *Ecosystems*, 14(2), 311–325. https://doi.org/10.1007/s10021-011-9411-4
- Tipping, E., Corbishley, H. T., Koprivnjak, J. F., Lapworth, D. J., Miller, M. P., Vincent, C. D., & Hamilton-Taylor, J. (2009). Quantification of natural DOM from UV absorption at two wavelengths. *Environmental Chemistry*, 6(6), 472–476. https://doi.org/10.1071/en09090
- Tucker, D. G., & Acreman, M. A. (2000). Modelling ditch water levels on the Pevensey Levels wetland, a lowland wet grassland wetland in East Sussex, UK. *Physics and Chemistry of the Earth, Part B: Hydrology, Oceans and Atmosphere*, 25(7–8), 593–597. https://doi. org/10.1016/s1464-1909(00)00070-8
- Vermaat, J. E., Hellmann, F., Dias, A. T., Hoorens, B., van Logtestijn, R. S., & Aerts, R. (2011). Greenhouse gas fluxes from Dutch peatland water bodies: Importance of the surrounding landscape. Wetlands, 31(3), 493–498. https://doi.org/10.1007/s13157-011-0170-y
- Von Arnold, K., Weslien, P., Nilsson, M., Svensson, B. H., & Klemedtsson, L. (2005). Fluxes of CO₂, CH₄ and N₂O from drained coniferous forests on organic soils. *Forest Ecology and Management*, 210(1–3), 239–254. https://doi.org/10.1016/j.foreco.2005.02.031
- Wallin, M. B., Audet, J., Peacock, M., Sahlée, E., & Winterdahl, M. (2020). Carbon dioxide dynamics in an agricultural headwater stream driven by hydrology and primary production. *Biogeosciences*, 17(9), 2487–2498. https://doi.org/10.5194/bg-17-2487-2020
- Wallin, M. B., Campeau, A., Audet, J., Bastviken, D., Bishop, K., Kokic, J., et al. (2018). Carbon dioxide and methane emissions of Swedish low-order streams—A national estimate and lessons learnt from more than a decade of observations. *Limnology and Oceanography Letters*, 3(3), 156–167. https://doi.org/10.1002/lol2.10061



- Wallin, M. B., Löfgren, S., Erlandsson, M., & Bishop, K. (2014). Representative regional sampling of carbon dioxide and methane concentrations in hemiboreal headwater streams reveal underestimates in less systematic approaches. *Global Biogeochemical Cycles*, 28(4), 465–479. https://doi.org/10.1002/2013gb004715
- Wallin, M. B., Weyhenmeyer, G. A., Bastviken, D., Chmiel, H. E., Peter, S., Sobek, S., & Klemedtsson, L. (2015). Temporal control on concentration, character, and export of dissolved organic carbon in two hemiboreal headwater streams draining contrasting catchments. *Journal of Geophysical Research: Biogeosciences*, 120(5), 832–846. https://doi.org/10.1002/2014jg002814
- Webb, J. R., Clough, T. J., & Quayle, W. C. (2021). A review of indirect N₂O emission factors from artificial agricultural waters. *Environmental Research Letters*, *16*, 043005. https://doi.org/10.1088/1748-9326/abed00
- Williams, P., Whitfield, M., Biggs, J., Bray, S., Fox, G., Nicolet, P., & Sear, D. (2004). Comparative biodiversity of rivers, streams, ditches and ponds in an agricultural landscape in Southern England. *Biological Conservation*, 115(2), 329–341. https://doi.org/10.1016/s0006-3207(03)00153-8
- Xiao, Q., Hu, Z., Hu, C., Islam, A. T., Bian, H., Chen, S., et al. (2021). A highly agricultural river network in Jurong Reservoir watershed as significant CO₂ and CH₄ sources. *Science of the Total Environment*, 769, 144558. https://doi.org/10.1016/j.scitotenv.2020.144558
- Zhou, Y., Xiao, Q., Yao, X., Zhang, Y., Zhang, M., Shi, K., et al. (2018). Accumulation of terrestrial dissolved organic matter potentially enhances dissolved methane levels in eutrophic Lake Taihu, China. *Environmental Science & Technology*, 52(18), 10297–10306. https:// doi.org/10.1021/acs.est.8b02163